Author’s response to reviews

Title: New Insight into the Informal Patients’ Payments on the Evidence of Literature: A systematic Review Study

Authors:
Arefeh Pourtaleb (arefehpourtaleb@yahoo.com)
Mehdi Jafari (mjafari@iums.ac.ir)
Hesam Seyedin (h.seyedin@yahoo.com)
Ali Akhavan Behbahani (aliakhavanbehbahani@gmail.com)

Version: 3 Date: 04 Oct 2019

Author’s response to reviews:

Dear Editor Assistant

Accept my Sincere Greetings

Thank you for accepting our manuscript “New Insight into the Informal Patients’ Payments on the Evidence of Literature: A systematic Review Study” and for giving us the opportunity to improve it. We would be very happy if the revised manuscript will now fulfil the high standards of the “BMC Health Services Research” for publication. According to your email, we have tried to address all the mentioned point precisely as outlined below:

- Ethics approval and consent to participate has been completed.
- The funding section has been completed.
- Supplementary files have been removed.
- Additional files section has been added after references section.
- All figures, tables, and supplementary/additional files have been checked to be cited within the text.
- Clean manuscript has been uploaded.
Do let us know if you have any further concerns. Thank you again for consideration.

Kind Regards

Arefeh Pourtaleb on behalf of authors

Email: arefehpourtaleb@yahoo.com